Author’s response to reviews

Title: Evidence-based medicine self-assessment, knowledge, and integration into daily practice: a survey among Romanian physicians and comparison between trainees and specialists

Authors:

Roxana-Denisa CAPRAȘ (capras.roxana@umfcluj.ro)
Adriana-Elena BULBOACĂ (adriana_bulboaca@yahoo.com)
Sorana Bolboacă (sbolboaca@umfcluj.ro)

Version: 3 Date: 07 Jan 2020

Dear Editors,

BMC Medical Education

Thank you for the e-mail sent on January 6, 2020 and for the good news.

Please find the revised manuscript. The following changes were done:
- a list of all the abbreviations used in the manuscript was included in the manuscript.
- the reference to Figures were uniformized as Fig. x.
- the references were checked and the formatings were uniformized.
- the apps, LE and DR abbreviations were define at the bottom of Table 3.

Thank you for your support.

Best regards,
Sorana D. Bolboaca